

Test suite assessment of safety-critical systems using safety tactics and fault-based mutation testing

Havva Gulay Gurbuz¹ · Bedir Tekinerdogan¹ · Cagatay Catal² · Nagehan Pala Er³

Received: 13 August 2023 / Revised: 28 November 2023 / Accepted: 29 November 2023 $\ensuremath{\mathbb{C}}$ The Author(s) 2024

Abstract

A safety-critical system is a system in which the software malfunctioning could result in death, injury, or damage to the environment. Addressing safety concerns early on at the architecture design level is critical to guide the subsequent life cycle activities to ensure that the eventual system is reliable. A fundamental approach to address safety at the design level is the adoption of architectural tactics. It is crucial for safety-critical systems to correctly implement the constraints as defined by the selected safety tactics. Given the limitations of traditional safety measures, particularly in conducting thorough analyses of safety tactics at the architectural design level, our motivation is to close this gap by integrating safety tactics and fault-based mutation testing. This article proposes a systematic approach for assessing the adequacy of test suites of safety-critical systems based on these architectural safety tactics. Our proposed approach involves the integration of safety tactics and fault-based mutation testing to comprehensively assess the adequacy of test suites in safety-critical systems. At the core of our methodology is the adoption of architectural safety tactics, addressing potential issues at the design level. We leverage fault-based testing, utilizing a dedicated metamodel and domain-specific language (DSL) to model safety views and their relation to the code. We use a case study to evaluate the effectiveness of our approach using fault-injection techniques. Our study shows that this systematic approach is feasible and effective for test suite assessment of safety-critical systems.

Keywords Software safety \cdot Safety-critical systems \cdot Fault-based testing \cdot Domain-specific language \cdot Test suite assessment \cdot Safety tactics

Bedir Tekinerdogan bedir.tekinerdogan@wur.nl

> Havva Gulay Gurbuz havva.gurbuz@wur.nl

Cagatay Catal ccatal@qu.edu.qa

Nagehan Pala Er npala@aselsan.com.tr

- ¹ Information Technology Group, Wageningen University & Research, Wageningen, The Netherlands
- ² Department of Computer Science and Engineering, Qatar University, Doha, Qatar
- ³ Aselsan, Ankara, Turkey

1 Introduction

Currently, an increasing number of safety-critical systems are controlled by software and rely on the correct operation of the software. Aircraft flight control, nuclear systems, medical devices are well-known examples of safety-critical systems. In this context, a safety-critical system is a system in which software malfunctioning could result in death, injury, or damage to the environment. The software can be considered as safe, which may not lead to a dangerous or life-threatening event for the system. In the literature several studies have discussed the methods, techniques, processes, tools, and models to make the software safe [1–3].

System safety engineering applies engineering and management principles, criteria, and techniques to optimize all aspects of safety within the constraints of operational effectiveness, time, and cost throughout all phases of the system life cycle [4, 5]. Software safety can be

addressed at different levels in the software development life cycle. Addressing safety concerns early on at the software architecture design level is crucial because quality characteristics such as safety cannot be included after the software implementation. An essential approach to address safety at the design level is the adoption of architectural tactics [6]. A tactic is a design decision for realizing quality goals at the architectural level. A safety tactic can be introduced for realizing safety. Wu and Kelly propose, for example, a set of tactics for software safety [7]. Based on the point at which faults are addressed for ensuring safety, we can categorize safety tactics as fault avoidance, fault detection, and fault tolerance safety tactics [8]. Safetycritical systems usually use a combination of these tactics to address the required safety concerns.

Once a safety-critical system is designed, it is crucial to analyze it for safety requirements before starting the implementation, installation, and operation phases. It is critical to ensure that the potential faults can be identified and cost-effective solutions are provided to avoid or recover from the failures. One of the most critical issues is investigating the effectiveness of the applied safety tactics to safety-critical systems. Several scenario-based software architecture analysis approaches [9-11] exist in the literature to analyze the architecture's quality. Unfortunately, these approaches are general purpose and do not directly consider safety concerns, thus fail to provide an in-depth analysis of the safety tactics.

In this article, we adopt a fault-based testing approach to analyze the effectiveness of the test suite of safety-critical systems using safety tactics. The novelties of this study are pointed out as follows: (1) building a systematic faultbased testing approach for assessing test suite adequacy based on architectural tactics and (2) developing Domain Specific Language (DSL) and tool for the proposed faultbased testing approach. An essential aspect of fault-based testing is mutation analysis which involves modifying a program under test to create mutants of the program. To apply fault-based testing for assessing the test suite, we first present a metamodel and a domain-specific language that models several safety views and their relation to the code. Mutants are generated for the potential hazards and the corresponding tactics. The approach results in the impact analysis of a test suite on the applied tactics. The proposed approach is illustrated using an industrial case study in the avionics domain. The case study demonstrates an important part of the aircraft control platform used in the developed avionics systems. With the case study, our tool allowed us to automate this process by removing manual steps for generating the mutations and running test cases. It also helped us ensure the safety concerns were properly addressed in the test cases by focusing on the safety tactics.

The remainder of the article is organized as follows. In Sect. 2, we present the required background information for understanding the overall approach. Section 3 provides a case study that we use to illustrate our fault-based testing approach. Section 4 presents the metamodel and domain-specific language for software safety tactics. In Sect. 5, we present the proposed fault-based testing approach. Section 6 presents our tool that implements the corresponding approach. In Sect. 7, we illustrate the proposed approach and the tool using the industrial case study. We present DSL evaluation in Sect. 8. Section 9 presents the related work. In Sect. 10, we present treats to validity and Sect. 11 concludes the paper.

2 Preliminaries

2.1 Safety tactics

Several studies [7, 12, 13] proposed architectural tactics or patterns for supporting safety design. Safety tactics are organized in [7, 13] based on fault avoidance, fault detection, and fault containment.

Fault avoidance aims to prevent faults from occurring in the system. Simplicity and Substitution are fault avoidance tactics. Fault detection focuses on monitoring the system and identifying faults when they occur in the system. Condition Monitoring, Sanity Check, and Comparison are tactics for fault detection. Fault containment seeks to limit the impact of the fault and prevent propagation of the fault. Fault containment includes Redundancy, Repair, Degradation, Voting, Override, and Barrier tactics. In this study, we refer to and reuse the tactics discussed in the literature [13]. Table 1 shows the safety tactics along with their descriptions.

2.2 Fault-based testing

Fault-based testing is one of the testing approaches which aims to analyze, evaluate, and design test suites by using fault data. Mutation testing is one of the common forms of fault-based testing. It aims to design new test cases by analyzing the quality of the existing test cases. Mutation testing involves modifying a program under test to create variants of the program. Variants are created by making small changes in the program following a pattern. Mutation operators are the patterns to change the program's code, and each variant of the program is called a mutant. A test suite is applied to both a mutant and the original program code. If the original code and mutant behave differently, the test suite can detect the change between the original and the mutant program. However, if the original code and mutant behave the same, the test suite is not adequate to detect the difference, and it needs to be improved.

Table 1 List of safety tactics

Safety tactic	Category	Description
Simplicity	Fault avoidance	Keep the system as simple as possible to avoid faults.
Substitution	Fault avoidance	Use more reliable components which are well-proven in safety domain to avoid faults.
Sanity check	Fault detection	Check whether a system state or value remains in a valid range defined in system specification.
Condition monitoring	Fault detection	Check whether a system value remains in a valid range compared to a more reliable reference value. Reference value is computed at run-time and it is based on system input values and is not pre-known value from the system specification.
Comparison	Fault detection	Compare the outputs of redundant systems to detect faults.
Diverse redundancy	Fault containment	Develop redundant components using different implementations based on the same system specification.
Replication Redundancy	Fault containment	Develop redundant system using the same implementation.
Repair	Fault containment	Bring a failed system back to its normal and healthy state and restore it.
Degradation	Fault containment	Brings a system with an error into a state with reduced functionality in which the system still maintains the core safety functions.
Voting	Fault containment	Mask the failure through choosing a correct result from redundant systems.
Override	Fault containment	Choose the output of redundant subsystems by preferring one subsystem or one output state over another.
Barrier	Fault containment	Protect a subsystem from influences or influencing other subsystems.

Mutation analysis consists of the following three steps [14]:

- 1. Mutation operator selection relevant to faults.
- 2. Mutant generation.
- 3. Distinguishing mutants by executing the original program and each generated mutants with the test cases.

After test cases are executed on mutated programs, the mutation score is calculated using the number of live mutants and the number of killed mutants. If the behavior/ output of a mutant differs from the original program, the mutant is killed. Otherwise, the mutant is alive. The mutation score is calculated using eq. (1). If a mutant's behavior is the same as the original program, the mutant is equivalent. Mutation score [15] is used to evaluate the adequacy of the test cases. The mutation score shows the effectiveness of test cases in terms of their ability to detect injected faults. A higher mutation score means a higher quality of test cases.

$$MutationScore = \frac{\# of \ killed \ mutants \times 100}{\# of \ total \ mutants - \# of \ equivalent \ mutants}$$
(1)

In the context of safety-critical systems, architectural-level mutation can be achieved by introducing mutations to the system's overall design, including changes to safety-critical components, alterations to communication protocols, and modifications to the system architecture. The goal is to assess how effectively the testing approach and test suite can identify and handle faults in the architectural design that may impact system safety.

3 Case study

This section describes a case study to illustrate our approach in subsequent sections. The case study is taken from an open-source software called Openpilot [16] implemented using Python and C++. Based on this study [17], it is one of the most popular open-source software in the safety-critical system domain. Openpilot is open-source driver assistance system developed by Comma.ai. It has Automated Lane Centering, Forward Collision Warning and Lane Departure Warning functionalities supporting a variety of car makes and models. It also has Driver Monitoring capability to alert distracted and asleep drivers. Openpilot consists of different components to communicate with the car and sensors, decide on the state of gas, brake, and steering, and process the sensor data to provide a safer driving experience for the drivers. Their high-level component diagram is given in [18]. In this study, we focus on Driver Monitoring capability which evaluates the data coming from sensors and generates alerts for drivers for a safer driving experience.



Fig. 1 High-level architectural diagram of the case study

In Fig. 1, we presented the high-level component diagram of Openpilot that we focused on this study. The overall component diagram for the Openpilot can be found in [18]. *AlertManager* is a module to process and manage the alerts. *Events* is a base module which defines *events* and *alerts* in Openpilot environment. *Controlsd* is a main module to combine wide range of inputs from sensors and car state and produces car-specific Controller Area Network (CAN) messages. CAN is a communication protocol. Electrical units and devices in the car communicate through CAN messages. *Controlsd* communicates with *AlertManager* to publish proper alerts to the user based on the inputs it receives.

For the case study, we select *displaying an alert to the* user when an unusual event occurs. As explained in [19], to ensure the safety of the driver, hazards and safety requirements should be identified and addressed accordingly. Hazard is a potentially dangerous situation that can result in or contribute to an accident [5]. For the case study, the hazard is displaying an incorrect alert or no alert to the *user*. The possible causes of this hazard are a loss of/error in the car sensors, a loss of/error in communication with the car sensors, an error in the display device, and an incorrect evaluation of the data coming from sensors. The possible consequence of this hazard is identified as a car accident. The severity of the hazard is catastrophic since the possible consequence of the hazard is a car crash. Based on this hazard, we define the safety requirements in Table 2.

Openpilot uses several safety tactics in their implementation to meet the defined safety requirements. In order to implemented SR5, they use Sanity Check tactic by checking the event type to decided whether it is an alert or not. If it is not an alert, they do not show it to the user. To implement SR6, they use Condition Monitoring and Sanity Check tactics to monitors the alerts' state and to decide which alert has a high priority to be shown to the user. Table 3 summarizes the rest of the applied tactics for the case study, along with the safety requirements.

4 DSL for safety

This section presents the metamodel and domain-specific language (DSL) for software safety to represent safety-related concepts. After a thorough domain analysis, our earlier work [8] derived a metamodel to express safety design concepts. In this work, we enhanced the earlier metamodel to support our fault-based testing approach. We updated the previous metamodel by adding *Architecture To Code Relation* part to show concepts and relations used in the fault-based testing approach to generate mutations and running test cases. We present the updated metamodel in Fig. 2.

The first part (Safety-Critical) of the metamodel includes the concepts present in the architecture design. Three types of architectural elements are distinguished as Monitoring Element, Safety-Critical Element, and Non-Safety Critical Element. Monitoring Element monitors one or more Safety-Critical Elements by checking their status. If there is a problem in a Safety-Critical Element, the Monitoring Element can react by stopping/starting/restarting/initializing the related Safety-Critical Element. Safety-Critical Element presents the element which includes safety-critical operations. A Safety-Critical Element can consist of one or more Safety-Critical Elements. We represented this relation in the figure using is element of. A Safety-Critical Element has States, including Safe State. If a fault is detected, which can lead to a hazard in the system and there is a safe state, the system can take itself to the safe state to prevent the hazard. In this regard, we have defined Safe State for defining safe states for Safety-Critical Elements. A Monitoring Element or Safety-Critical Element applies Safety Tactics in order to ensure the safety of the system.

The second part of the metamodel includes the concepts related to applied safety tactics in the design. We have identified the well-known safety tactics as fault avoidance, fault detection, and fault tolerance. Fault avoidance tactic aims to prevent faults from occurring in the system. When a fault occurs, the fault is detected by applying fault detection tactics. Fault tolerance is the ability of the system to continue correctly and maintain a safe operational condition when a fault occurs. Therefore, applied *Safety Tactic* can be *Fault Avoidance Tactic*, *Fault Detection Tactic*, or *Fault Tolerance Tactic* to deal with faults.

The third part of the metamodel includes the concepts which are related to hazards in the system. A *Hazard* describes the presence of a potential risk situation that can result or contribute to the mishap. A *Hazard* causes some *Consequences. Safety Requirements* are derived from identified *Hazards*. For the safety-critical systems, a thorough hazard analysis should be done to discover potential

Table 2 Safety requirements	for the selected hazards
-------------------------------------	--------------------------

Safety requirement	Explanation
SR1	Events should be evaluated at least from two different components.
SR2	If only one of the components produces an event, the incoming event should be evaluated and a warning should be generated.
SR3	If both components cannot produce an event, the error should be generated.
SR4	The two events should be compared and if they are not same, always the event coming from the selected source should be displayed along with a warning.
SR5	If an event is not an alert, do not show it as an alert.
SR6	If there are multiple alerts that occurring at the same time, show the most recent and high prioritized event as a current alert.
SR7	If an alert is not active, even if it is the most recent and high prioritized one do not show it as a current alert.

Table 3 Applied safety tactics to case study

Safety req.	Safety tactic and category	Tactic description
SR1	Replication redundancy (fault containment)	The events are evaluated by two different Alert Managers where both has the same logic to evaluate events.
SR 2	Condition monitoring (fault detection) voting (fault containment)	The health of both Alert Managers should be periodically monitored to see whether they are healthy or not. If one of them is failing, the event from the other manager will be displayed.
SR3	Condition monitoring (fault detection) repair (fault containment)	The health of both Alert Managers should be periodically monitored to see whether they are healthy or not. If there is a failure on any of the managers, they will be put in repair mode. If both of them are failing, an error will be generated and no alert will be shown.
SR 4	Comparison (fault detection) override (fault containment)	If the events produced by each Alert Manager are not same, always show the event coming from Alert Manager 1 and display a warning.
SR5	Sanity check (fault detection)	The given event is checked if it is meets the criteria for representing an alert in the Openpilot environment.
SR6	Sanity check (fault detection) comparison (fault detection)	All the existing alerts' states are monitored and validated against the pre-defined criteria to decide which alert is going to be shown to the user.
SR7	Sanity check (fault containment)	Monitor the alert's state and if it is not active do not show it to the user.

hazards and identify their root causes. Fault Tree Analysis [1] is one of the most well-known and widely used methods for hazard analysis. It aims to analyze a design for possible faults that could lead to hazards in the system using Boolean logic. We define *FTA Node*, *Operator*, and *Fault* to conducting Fault Tree Analysis. *FTA Nodes*, *Faults*, and *Operators* are the elements of a fault tree. *Operator* is used to conduct Boolean logic. Operator can be AND or OR. One or more *FTA Nodes* cause a *Hazard*.

The last part of the metamodel is *ArchitectureToCodeRelations* defined in the fault-based testing process for mutant generation and test case run steps. As presented in Fig. 2, *ArchitectureToCodeRelations* consists of Implementation Relations which can be *Module-Class* Relation or Class-Test Case Relation. Module-Class Relation describes which Safety-Critical Elements defined in Safety-Critical View consists of which classes in the program code. Class-Test Case Relation defines which classes in the program code should be tested with which test cases. Based on the safety metamodel presented in Fig. 2 we provide a domain-specific language (DSL) to represent the concepts in the safety domain. The EBNF grammar [20] of this DSL is presented in 12... In the code snipped below, we present an example definition of a hazard using our DSL. It illustrates the hazards "displaying an incorrect alert or no alert to the user" and "violating maintaining a safe distance" from the case study that we explained in Sect. 3.

```
HazardView Openpilot_HazardView{
    Elements {
        hazard displayingAnIncorrectAlertToUser;
        safetyRequirement doNotDisplayNonAlertEvents;
        safetyRequirement showHighPriAndMostRecentAlert;
        safetyRequirement doNotShowExpiredAlerts;
        consequence carAccident;
    }
    Relations {
        doNotDisplayNonAlertEvents,
        showHighPriAndMostRecentAlert,
        doNotShowExpiredAlerts derivedFrom displayingAnIncorrectAlertToUser;
        displayingAnIncorrectAlertToUser causedBy incorrectAlert;
    }
}
```

5 Fault-based testing approach

In the previous sections, we describe the metamodel and the corresponding DSL for modeling safety-critical architectural concerns and the relation to the implementation. Following up on this, Fig. 3 shows the process of our faultbased testing approach.

Our fault-based testing (FBT) approach leverages the DSL, and we use mutation testing to evaluate the test suite's quality. Within our approach, we enhanced the classical method for mutation testing with

- providing a guideline for how to select and decide on mutation operators for applied safety tactics
- automating the mutation generation (for implementation-level mutations), and test case execution
- defining a scope for the testing process by only focusing the safety requirements and the safety tactics

The approach consists of several steps: identifying safety requirements and safety tactics, building a safety model and mutation model, generating mutants, running the test suite on the generated mutants, and evaluating the results. To build a safety model and mutation model, we first need to define the safety concerns in the system. For this, we start our process by identifying safety requirements and safety tactics. In the following, we explain our proposed approach in detail.



Fig. 2 Metamodel for safety DSL



Fig. 3 Process of proposed fault-based testing approach

While the time and effort required for each step can vary based on the complexity of the safety-critical system and the specific project requirements, we can provide a general estimation based on our experience. Conducting hazard analysis may take approximately 8 h, involving a team effort of 2 engineers for 4 person-hours each. The process of deriving safety requirements might require around 6 h and a team effort of 1 safety engineer and 1 domain expert, each contributing 3 person-hours. Manual creation of safety models could take approximately 4 h, with a team effort of 2 engineers contributing 2 person-hours each.

5.1 Identifying safety requirements

The first step of our proposed approach is identifying the safety requirements of the system. Safety requirements are defined based on the hazards and risks in the system [21]. Hazard analysis is performed to identify the hazards in the system by building a list of all hazards, their causes, consequences, and severity. Hazard severity levels are defined as catastrophic, critical, marginal, or negligible in [21]. Hazard identification activity is performed with domain experts (avionics engineers and pilots), system engineers, and safety engineers. The risks in the system are defined by the estimation of the probability of occurrence of each hazard. In [21], occurrence definitions are classified as frequent, probable, occasional, remote, or improbable. Based on the hazard severity and hazard occurrence class identification, risks should be assessed and categorized as high, serious, medium, or low. After the risk definition, a risk assessment should be conducted using fault tree analysis, event tree analysis, simulation, etc. Safety requirements can be derived using identified hazards and risks. In the following subsections, we use "maintaining a safe distance with the leading car" as an example safety requirement. For this requirement, a hazard would be "failing to maintain a safe distance on autopilot mode for autonomous driving cars".

5.2 Identifying safety tactics

As a second step, safety tactics should be defined to satisfy identified safety requirements. In Table 1, we provided the list of well-known safety tactics that can address safety concerns. Safety tactic(s) should be determined to avoid failures and hazards for each identified safety requirement. Table 1 can be leveraged as a guideline to determine safety tactics for the hazards defined in the identifying safety requirements step. Based on the example we defined Section in 5.1, below are the example safety tactics that can be defined to avoid this hazard using Table 1:

- Sanity Check: Check the distance with the leading car and ensure that it stays within the defined threshold distance.
- Diverse redundancy: Calculate the distance with the leading car from at least two different software components to reduce the risk of miscalculation.

5.3 Building safety model

The next step is creating a safety model using the safety DSL. We use the safety model to generate mutants and run test cases. Hazard view, safety tactic view, and safety-critical view should be defined in order to construct a safety model.

5.3.1 Hazard view

The hazard view should include the safety requirements and the hazards derived from the safety requirements in the first step of our FBT approach. In addition to hazards, the model should contain failures and faults that the identified hazards can cause. The code snippet below shows a simple hazard view for the example hazard we have defined in the previous subsections.

```
HazardView HazardViewExample{
    Elements {
        hazard failingToMaintainSafeDistance;
        safetyRequirement maintainSafeDistance;
        consequence carAccident;
        fault lossOfCarSensor;
        fault errorInCarSensor;
        // more faults ...
        faultTree incorrectDistanceCalculation(
         // fault tree definition
        );
    }
    Relations {
        maintainSafeDistance derivedFrom failingToMaintainSafeDistance;
        failingToMaintainSafeDistance causedBy incorrectDistanceCalculation;
    }
}
```

5.3.2 Safety tactic view

The safety tactic view consists of the safety tactics identified in the second step of our FBT approach. The safety tactic view should contain the information on "Sanity Check" and "Diverse Redundancy" tactics we defined in Sect. 5.2. The code snippet below shows a simple safety tactic view for the example scenario we have defined.

5.3.4 Implementation relations view

For the mutant generation and test case execution steps, architecture to code relations should also be defined. The code snippet below shows a simple architecture to code relationships view for the example scenario we have defined. In this example, with module-class relations, we

```
SafetyTacticView SafetyTacticsViewExample{
   faultContainment redundantDistanceCalculator {
     type = "DiverseRedundancy"
     containedFaults = // faults
   };
   faultDetection safeDistanceCheck {
        type = "SanityCheck"
        detectedFaults = // faults
   };
}
```

5.3.3 Safety-critical view

The safety-critical view describes the architectural components of the system from a safety perspective. The code snippet below shows a simple safety-critical view for the example scenario we have defined. In this example, since in the system we apply diverse redundancy tactic, we have two distinct components to calculate the distance between the leading car. This view also includes other safety-critical, non-safety-critical and other elements in the system. indicate that the "distanceCalculatorComponent1" includes "Distance" and "CalculatorComponentA" implementation classes/files. And with class-test case relations, we indicate that the tests for "Distance" class/file lives in test suite "distanceTests" where the tests for "CalculatorComponentA" lives in test suites "calculatorTests" and "componentATests".

```
SafetyCriticalView SafetyCriticalViewExample{
    Elements {
        safetyCritical distanceCalculatorComponent1{
            criticalityLevel = B;
            implementedSafetyRequirements = maintainSafeDistance;
            implementedTactics = redundantDistanceCalculator;
        };
        safetyCritical distanceCalculatorComponent2{
            criticalityLevel = B;
            implementedSafetyRequirements = maintainSafeDistance;
            implementedTactics = redundantDistanceCalculator;
        1:
        // other elements
    3
    Relations {
        // elements relations
    }
}
```

```
ImplementationRelations {
    ModuleClassRelations {
        distanceCalculatorComponent1 composesOf = {Distance, CalculatorComponentA };
        distanceCalculatorComponent2 composesOf = {Distance, CalculatorComponentB };
    };
    ClassTestCaseRelations {
        Distance testWith = {distanceTests};
        CalculatorComponentA testWith = {calculatorTests, componentATests};
        CalculatorComponentB testWith = {calculatorTests, componentBTests};
    };
}
```

5.4 Mutant generation

In order to generate mutants, we need to know what part of the system needs to be changed and how its behavior is going to be changed. With safety models, we already define what part of the system needs to be changed by focusing on safety tactics and the safety-critical components in the system. We introduce a mutation model for each safety tactic to define the behavioral change. Each mutation model describes the possible ways of changing the behavior of the applied safety tactic. Using these models, the mutation generation can be achieved. Each row in Table 4, explains a mutation model for well-known safety tactics listed in Table 1. Each mutation model is defined based on the tactic properties column. If the safety tactic is addressed on the implementation (code) level, the mutation is also applied on implementation-level. Where if the safety tactic is addressed on the component level, the mutation is applied at architectural-level. Table 4 also includes common mutation operators related to each implementation-level mutation model. If we take the example safety tactics we defined in Sect. 5.2, Sanity Check tactic requires having a range check on a system state or value to check their validity. The mutation model for this tactic would be on the implementation-level where we mutate the tactic implementation by adding or removing the arithmetic, relational and conditional operators. If we consider Diverse Redundancy tactic, it requires having redundant components which they have different implementations. The mutation model for this tactic would be on the architectural-level where we add redundant components to have each component have the same implementation. Table 4 can be used as a guideline to build mutation models for each safety tactic defined for the given safetycritical system. The relations between safety tactics, mutation models and mutation operators in Table 4 are addressed in our tool which is presented in Sect. 6.

There are several tools in the literature to generate implementation-level mutants. MutPy is [22] is one of the mutation testing tool for Python 3.3+ for generating implementation-level mutants automatically. We use Mut-Pv's guideline while selecting the proper mutation operators for the safety tactics that we applied to our case study. We have a code generation process where it provides an automated way for creating implementation-level mutations. The code generation process uses the mapping between safety tactic and MutPy mutation operator presented in Table 4. For the code generation process, we have leveraged the code generator provided by Xtext framework [23]. Xtend [24] is part of the Xtext framework, and it is used for model-to-model or model-to-text transformation. We used Xtend to generate a code from our safety model (model-to-text transformation). The code generator is part of the tool we developed which is presented in Sect. 6. For the code generation process, we need the parts of the program code are going to be mutated and what type of mutation is going to be applied. The code generation process extract this information from the safety model of the system. Below are the steps for generating implementationlevel mutations within our tool.

- 1. Find applied safety tactic(s) for each safety requirement using the hazard and safety tactic view. Create a mapping between safety requirement and associated safety tactics.
- 2. Find safety-critical modules for each safety tactic using the safety-critical view and create a mapping between them.
- 3. Find implementation files/classes for each safetycritical modules from implementation relations view and create mapping between them.
- 4. Find test suites for each implementation file/class from implementation relations view and create mapping between them.

 Table 4 Mutation model for safety tactics

Safety tactic	Tactic properties	Mutation model
Simplicity	N/A	N/A
Substitution	- Using more reliable components	- Architectural-level mutation: Replace the components with faulty components
Sanity check	- Checking the system state or value to see if it remains within a valid range determined in the system specification	- Implementation-level mutation: Mutate the implementation of checking logic by replacing, adding or removing the arithmetic, relational and logical operators Mutation operators: Arithmetic operators (+, -, *, /, //, ^, %, unary + and unary -), Logical operators (&&, , !), Relational operators (<, \leq , >, \geq , !=, ==)
Condition monitoring	- Checking the system state or value to see if it remains within a valid range calculated at run time - Having a component for monitoring	- Mutate the implementation of checking logic by replacing, adding or removing the arithmetic, relational and logical operators Mutation operators: Arithmetic operators (+, -, *, /, //, ^, %, unary + and unary -), Logical operators (&&, , !), Relational operators (<, \leq , >, \geq , !=, ==) - Architectural-level mutation: Replace the monitoring component with the faulty one
Comparison	- Comparing the values from redundant components	- Implementation-level mutation: Mutate the implementation of comparison logic by replacing, adding, removing the relational and logical operators Mutation operators: Logical operators (&&, , !), Relational operators ($<, \le, >, \ge, ! =, ==$)
Diverse redundancy	- Having different implementations for redundant components or subsystems	- Architectural-level mutation: Replace the redundant component(s) with a component which has a same implementation of the primary component.
Replication redundancy	- Having same implementation for redundant components or subsystems	- Architectural-level mutation: Replace the redundant component(s) with a component which has different implementation from the primary component.
Repair	- Having a component for repairing the failed components	- Architectural-level mutation: Replace the implementation of a repairing component with a component that has missing functionalities or a faulty one
Degradation	- Having a component that brings the system in a state with reduced functionalities but has the core safety functions	- Architectural-level mutation: Replace the implementation of a degradation component with a component that has missing functionalities or the faulty one
Voting	- Having a component that chooses the majority of the output values as output	- Implementation-level mutation: Replace the implementation of the voting component as to select output value randomly. There is no mutation operators suggested for this, this requires a domain knowledge of the system to mutate the program code.
Override	- Having a component that chooses the output of redundant components by preferring one output over another	- Implementation-level mutation: Replace the overriding logic by selecting the output value from one of the components other than the preferred one. There is no mutation operators suggested for this, this requires a domain knowledge of the system to mutate the program code.
Barrier	- Having a barrier component for protecting a component from influences or influencing other components	- Architectural-level mutation: Replace the barrier component with one that allows the components to have an effect on each other.

Pseudo code for mutant generation:

The code generation process uses the mapping between safety tactics and mutation models defined to decide on the mutation behavior. The mutation model tells us what type of mutations are going to be applied. If the mutation is at architectural-level, the mutation generation needs to be performed manually. On the other hand, for the implementation-level mutations, our tool generates a code snippet with the mutation operators defined in Table 4 so that mutants can be generated. In order to get the parts of code to be mutated for implementation-level mutations, our tool extracts the module-class relations and test classes-class relations from Architecture To Code Relations for the safety-critical elements we have obtained. These relations help us to get the implementation details such as module, the class, and the test class. This information indicates the parts of the code are going to be mutated. For each safety tactic in the system, Python code is generated using the extracted information. The generated code is skeleton code which has the required code to generate mutants and run test cases by executing related methods from *MutPy*. We provide the mutation operators to the skeleton code and run the complete code with the original program code. Each selected mutation operator is switched with the operator in the original code by MutPy, and mutants are generated. Architectural-level mutation requires adding, removing, or modifying a component. Since it requires implementation-specific knowledge, it can be achieved by updating the component manually, or this process can be fully or partially automated depending on the case.

Considering the example scenario we provided in previous sections, the mutation model for the SanityCheck tactic is on the implementation level and it requires replacing, adding or removing the arithmetic, relational and logical operators in the program code.

5.5 Running test cases on the mutants and mutation score evaluation

When we have the mutants generated, as a next step, the test cases are run on the mutants to assess the quality of the test suite. Our study focuses on evaluating the quality of the existing test suite. From this perspective, we use the test suite implemented during the system development. Our approach does not include a process for generating test cases. For implementation-level mutations, this step is also automated. Test cases are run by executing the generated code. For the architectural-level mutations, we run the test cases manually. Test suite evaluation is performed on the implementation level.

Based on the results of the test case execution step, we calculate the mutation score and evaluate it. If there is an alive mutant (not killed by any test cases), we add new test cases to handle the alive mutants. This process is repeated until all the mutants are killed.

6 Tool

In this section, we present the tool that we developed (i.e., Safety DSL) [25] in the Eclipse environment to define safety models using safety DSL and the Python script to apply a fault-based testing process.

We defined the grammar of safety DSL using Xtext [23], a language development framework provided as an Eclipse plug-in. After defining our DSL in Xtext, we wrote our code generator using Xtend provided in Xtext framework for the safety DSL. Xtext and the corresponding code generator create the parser and runnable language artifacts. From these artifacts, Xtext generates a full-featured Eclipse text editor. Figure 4 shows the snapshot of the Eclipse text editor for our case study. As explained in the previous section, for the mutant generation and test case execution steps, an existing open-source Python project MutPy is used. MutPy provides mutation operators for the mutant generation. Additionally, it enables to execution of predefined test cases on mutated program code. The Python script is generated during the code generation process to mutate the program code and execute test cases leveraging MutPy.

7 Case study evaluation

This section explains the application of our fault-based testing approach and presents the results by using an industrial case study described above. We applied the

runtime-Eclipse	(text - safetyDSL/src/openPilotAlertManager.safety - Ecl	ipse Platform	
- 🗂 • 🗐 🐚 🔌 🔅 • 🕥 • 🂁 • 🔐 🧿	ŷ• 🤔 ⋵ - ৵• 월• 禄• फ फ • ↔ 🥖	Quick Access	🗈 🗗 🖏
ቹ Package Explorer 🔀 🛛 🗖 🗖	B openPilotAlertManager.safety ☎	- 8	т¤ □ □
 E SeifetyDSL > ▲ JRE System Library [JavaSE-' > ● src > ▷ src-gen 	<pre> Hazard View Openpilot_HazardView{ Elements { hazard displayingAnIncorrectAlertToUs safetyRequirement evaluateEventsWithT safetyRequirement oneEventCannotBePro safetyRequirement bothEventsCannotBeP safetyRequirement doNotDisplayNonAler safetyRequirement showHighPriAndMostR safetyRequirement doNotShowExpiredAle </pre>	er; woComponents; duced; roduced; nt; tEvents; ecentAlert; rts;	
	consequence carAccident; fault lossOfCarSensor;		• ▼ □ □
twoEventsAreDifferent twoEventsAreDifferent doNotDisplayNonAlertEven showHighPriAndMostRecer sdoNotShowExpiredAlerts carAccident lossOfCarSensor serrorInCarSensor	0 items De Resource	Path	Location
Writable	Insert 16:32		0

Fig. 4 Tool for safety DSL

process shown in Fig. 3. In the following, we explain the application of each step.

7.1 Build Safety Model

As described in Sect. 3, we selected *displaying an alert to user when an unusual event occurs* hazard and identified safety requirements as shown in Table 2. Later, we defined the corresponding safety tactics and presented them in Table 3. In order to address the safety tactics we defined, we leveraged the part of Openpilot open-source software explained in Sect. 3 and built a case study. Figure 5 shows the overall high-level architecture diagram of our case study. We added another AlertManager (AlertManager2) to address Replication Redundancy defined in 3 for SR1. Table 5 presents the links to the Openpilot module associated with the component shown in Fig. 5.

These are the first two steps of the proposed fault-based testing approach. As a next step, we built the safety model. For defining the safety-critical view, firstly, we identified our architectural elements. AlertManager1 and AlertManager2 shown in Fig. 5 are responsible for processing alerts. Each AlertManager receives the alert data from Controls. Controls read the alert data from car sensors using CAN protocol. If a warning or should be generated, AlertManagers notifies the Controls through commands relation. If a fault occurs in AlertManager1 and AlertManager2, they report the fault to Controls through reportsFault relation.



Fig. 5 High-level architectural diagram of the case study

For condition monitoring, voting, and recovery tactics, we added AlertManagerMonitor. AlertManagerMonitor monitors AlertManager1 and AlertManager2 components. It detects the failure when one of these managers fails and recovers from failures by stopping/starting/initializing the failed modules.

We built a safety model by using safety DSL explained in Section 4 according to the case description. We defined hazard view, safety tactic view, and safety-critical view. For the sake of simplicity, we present a small part of the code snippet from the safety-critical view below. Hazard view and safety-tactic view are created similarly using the safety DSL.

```
SafetyCriticalView Openpilot_Safety_Critical{
          Elements {
                   monitor alertManagerMonitor {
                             implementedTactics = healthCheckForAlertManager, recoverAlertManager
                   }:
                    safetyCritical alertManager1{
                             criticalityLevel = B;
                             implementedSafetyRequirements = showHighPriAndMostRecentAlert,
                                       evaluateEventsWithTwoComponents;
                             implementedTactics = checkAlertStates, compareAlertsTimeAndPriority,
                                       alertManagerReplica;
                   };
                    safetyCritical alertManager2{
                             criticalityLevel = B;
                             implementedSafetyRequirements = showHighPriAndMostRecentAlert,
                                       evaluateEventsWithTwoComponents:
                             implementedTactics = checkAlertStates, compareAlertsTimeAndPriority,
                                      alertManagerReplica;
                   };
                    safetyCritical controlsd{
                             criticalityLevel = B:
                             implementedSafetyRequirements = doNotShowExpiredAlerts,
                                       \verb+oneEventCannotBeProduced, \verb+bothEventsCannotBeProduced, \verb+twoEventsAreDifferent+ \verb+oneEventsAreDifferent+ $+ oneEventsAreDifferent+ $+ oneEventsAreDifferen+ $+ oneEvent+ $
                             implementedTactics = checkIfAlertIsActive, alertManagerSelectAlert,
                                       compareAlerts, alertManagerVoting;
                   };
                    safetyCritical events{
                             criticalityLevel = B;
                             implementedSafetyRequirements = doNotDisplayNonAlertEvents;
                             implementedTactics = eventTypeCheck;
                   };
         }
          Relations {
                    alertManagerMonitor monitors alertManager1, alertManager2;
                   alertManagerMonitor stops alertManager1, alertManager2;
                   alertManagerMonitor starts alertManager1, alertManager2;
                   alertManagerMonitor inits alertManager1, alertManager2;
                   controlsd reads alertManager1, alertManager2;
                   controlsd commands alertManager1, alertManager2;
                   alertManager1 reportsFault controlsd;
                   alertManager2 reportsFault controlsd;
         }
}
```

 Table 5
 High-level architectural component along with Openpilot module

Component	Openpilot Mmodule - Github Link
Events	events.py
Alert Manager1	alertmanager1.py
Alert Manager2	alertmanager2.py
Alert Manager monitor	alertmanagermonitor.py
Controls	controlsd.py

Additionally, we defined implementation relations for the mutant generation and test case run steps. Module-Class Relations shows which safety-critical module consists of which implementation classes, Class-Test Case Relations shows which implementation class should be tested with which test case classes. We present the architecture to code relations in the code snippet below. Architecture to code relations provides a mapping between the module defined in the architectural model and the class implemented in Java. Also, they map the implementation classes and test classes. For example, the code snippet shows that *alertManager1* has the implementation in *alertManager.py* in module-class relations. In the next section, class-test case relations show that the test cases for *alertManager.py* is implemented in *test_alerts.py*, *test_state_machine.py*, *test_alertmanager.py*. The complete safety model for our case study can be found in [25]. This code snippet includes the mutant generation code for the *Sanity Check* tactic for the module Controls. The mutation operators that we have in the above code snippet are AOD (Arithmetic Operator Deletion), AOR (Arithmetic Operator Replacement), COI (Conditional Operator Insertion), COD (Conditional Operator Deletion), ROR

```
ImplementationRelations {
   ModuleClassRelations {
       alertManagerMonitor composesOf = { selfdrive.controls.lib.alertmanagermonitor };
       alertManager1 composesOf = { selfdrive.controls.lib.alertmanager };
       alertManager2 composesOf = { selfdrive.controls.lib.alertmanager2 };
       events composesOf = {selfdrive.controls.lib.events};
       controlsd composesOf = { selfdrive.controls.controlsd };
   };
    ClassTestCaseRelations {
        selfdrive.controls.lib.alertmanagermonitor testWith = { selfdrive.controls.lib.
            tests.test_alertmanagermonitor };
       selfdrive.controls.lib.alertmanager testWith = { selfdrive.controls.tests.
            test_alerts, selfdrive.controls.tests.test_state_machine, selfdrive.controls.
            lib.tests.test_alertmanager };
        selfdrive.controls.lib.alertmanager2 testWith = { selfdrive.controls.tests.
            test_alerts, selfdrive.controls.tests.test_state_machine, selfdrive.controls.
            lib.tests.test_alertmanager };
       selfdrive.controls.lib.events testWith = { selfdrive.controls.lib.tests.
            test_alertmanager };
        selfdrive.controls.controlsd testWith = { selfdrive.controls.tests.
            test_state_machine };
   };
```

7.2 Identify and create mutants

In this step, we identified mutants based on the safety tactics that we implemented in our case study. Based on the mutation model we introduced in Table 4, we determined the mutants and presented them in Table 6. Table 4 explains the action item required to taken for each safety tactic. For example, for SR1, we have Replication Redundancy as a safety tactic in which we have AlertManager1 and AlertManager2 components as an application of this tactic. Based on the guideline we have in 4 we defined specific mutation models for each safety tactic we defined for the safety requirements.

We used the safety model and the selected mutation operators as inputs to our tool to create skeleton code for generating mutants for method-level mutation generation. The skeleton code includes the required Python code for mutant generation and execution of test cases. A sample code snippet is shown in below: (Relational Operator Replacement). LOR (Logical Operator Replacement), LOD (Logical Operator Deletion). COI and COD are works with conditional operators like && (and), \parallel (or), & (bit-wise and), \mid (bit-wise or), (xor), ! (not) where ROR is related to relational operators < (greater than), > (less than), < = (greater than or equals), >= (less than or equals), == (equals), != (not equals). All of the details on these operators can be found in [22].

For component-level mutation generation, we manually modified the code parts in the implementation of the case study to reflect mutations.

7.3 Run test cases

The next step is executing test cases on mutant codes. Test case generation is performed by generated code for method-level mutants. As shown in Implementation Relations, *controlsd.py* should be tested with *test_state_machine* test class. Based on the generated code, test cases are

Table 6 Identifying Mutant Model for Case Study

Safetyrequirement	Safety tactic	Mutation model
SR1	Replication redundancy	Component level: Replace the implementation of alert managers such that they have different implementations.
SR2	Condition monitoring voting	Method level: Use mutation tool to generate mutations of corresponding code part for AlertManagerMonitor Component level: Create the mutations of the AlertManagerMonitor as to have faulty voting behavior
SR3	Condition monitoring repair	Method level: Use mutation tool to generate mutations of corresponding code part for AlertManagerMonitor Component level: Create the mutations of the AlertManagerMonitor as to have faulty repair behavior
SR4	Condition monitoring override	Method level: Use mutation tool to generate mutations of corresponding code part for Controls Component level: Create the mutations of the Controls as to have faulty override behavior
SR5	Sanity check	Method level: Use mutation tool to generate mutations of corresponding code part for Events
SR6	Sanity check comparison	Method level: Use mutation tool to generate mutations of corresponding code part for Alert Managers
SR7	Sanity check	Method level: Use mutation tool to generate mutations of corresponding code part for Controls

Table 7 Mutation Results for SR1

	Lines of mutatedcode	Total testcases	Mutation model	Total generatedmutants	Equivalentmutants	Alivemutants	Killedmutants	Mutation score(%)
Alert Manager1	37	3	Replication redundancy	4	0	0	4	100
Alert Manager2	37	3	Replication redundancy	4	0	0	4	100

executed on mutants for each test class, and the results are then collected to generate a report.

7.4 Results

The last step is the generation of the report. When the test cases are executed in Sect. 7.3, results are collected, and the code part is called to generate a report. The report includes the classes under test, test case classes, mutation operators, test results (fail/pass), related faults, and related safety tactics. Tables 7, 8, 9, 10, 11, 12, and 13 show the results for our case study along with the mutation score. They also include details of mutant generation (the total number of lines of the mutated code, mutation model, the total number of generated mutants, the total number of alive mutants, and the total number of killed mutants). While calculating the mutation score, equivalent mutants should be determined. If a mutant semantically behaves precisely like the original program, the mutant is equivalent. We manually checked the generated mutants to see if they behave like the original code to detect equivalent mutants. This process can be improved by using the existing approaches proposed in several studies, such as [26–28]. Since the original program passes all of the test cases and the killed mutant is a mutant that failed on at least one of the test cases, a killed mutant cannot be an equivalent mutant. In this regard, we only checked the live mutants to see if any of them is an equivalent mutant. We included the number of equivalent mutants as another column in the tables. We calculated the mutation score using the formula presented in Sect. 5.

Tables 7, 10, 11 and 13 show the results for SR1, SR4, SR5 and SR7 respectively. For all of these, the mutation score is 100, no further action is required.

Table 8 presents the mutation results for SR2. For *Condition Monitoring and Voting* tactic, there are 10 mutants generated and 8 of them killed. When we revisited the test cases for the Condition Monitoring tactic, we observed that some cases were not considered for this tactic. The test cases were missing some of the edge cases for checking the state of each AlertManager. We added three more test cases to cover all of the cases. With the complete test suite, all of the mutants were killed, and we obtained a mutation score of 100.

Table 9 presents the mutation results for SR3. For Condition Monitoring and Repair tactic, there are 11

Table 8 Mutation results for SR2

	Lines of mutated code	Total test cases	Mutation model	Total generated mutants	Equivalent mutants	Alive mutants	Killed mutants	Mutation score (%)
Alert Manager Monitor	12	5	Condition monitoring and voting	10	0	2	8	80

Table 9 Mutation Results for SR3

	Lines of mutated code	Total test cases	Mutation model	Total Generated mutants	Equivalent mutants	Alive mutants	Killed mutants	Mutation score (%)
Alert Manager Monitor	19	4	Condition Monitoring and Repair	11	2	6	5	55.55

Table 10 Mutation Results for SR4

	Lines of mutated code	Total test cases	Mutation model	Total generated mutants	Equivalent mutants	Alive mutants	Killed mutants	Mutation score (%)
Controls	247	8	Condition monitoring and override	158	0	0	158	100

Table 11 Mutation results for SR5

	Lines of mutated code	Total test cases	Mutation model	Total generated mutants	Equivalent mutants	Alive mutants	Killed mutants	Mutation score (%)
Events	13	5	Sanity check	45	0	0	45	100

Table 12 Mutation Results for SR6

	Lines of mutated code	Total test cases	Mutation model	Total generated mutants	Equivalent mutants	Alive mutants	Killed mutants	Mutation score (%)
Alert Manager1	37	5	Sanity check comparison	21	1	14	6	30
Alert Manager2	37	5	Sanity check comparison	21	1	14	6	30

Table 13 Mutation results for SR7

Lines of Mutated Code	Total test cases	Mutation model	Total generated mutants	Equivalent mutants	Alive mutants	Killed mutants	Mutation score (%)	
Controls	342	8	Sanity check	258	0	0	258	100

mutants generated and 5 of them killed. When we revisited the test cases for the Condition Monitoring tactic, we observed that some cases were not considered for this tactic. The test cases were missing some of the edge cases for checking the state of each AlertManager. We added two more test cases to cover all of the cases. With the complete test suite, all of the mutants were killed, and we obtained a mutation score of 100.

Table 12 presents the mutation results for SR6. For *Sanity Check and Repair* tactic, there are 11 mutants generated and 5 of them killed. When we revisited the test cases for Repair tactic, we observed that some cases are not considered for this tactic. The test cases were missing some of the edge cases for checking the state of each AlertManager. In first iteration, we added two more test cases to cover all of the cases. With the first iteration, we were able to kill 5 more mutants in which the mutation score is 55%. In the second iteration we added two more test cases and we were able to obtain the mutation score as 100%.

7.5 Evaluation

In this section, we present the evaluation of our fault-based testing approach. We generated mutations for the implementation of the case study using the proposed approach and aim to achieve a 100% mutation score to locate the weaknesses in the test suite and have effective tests for safety concerns. For some of the safety requirements, the mutation score was already 100%. For these kinds of requirements, the implemented test suite is able to cover all the edge cases. However, for example, SR6 requires multiple safety tactics to be implemented, and these tactics have lots of edge cases to check. We revisited the test cases and observed that the test suite was missing some test cases.

In our evaluation, we specifically focused on the relevance of the mutation-based testing approach to safety requirements, with a dedicated examination of the Openpilot case study. The iterative nature of our approach ensured a thorough consideration of safety concerns within the test cases. Automation, facilitated by our tool in mutation generation and test case execution, streamlined this process. Our fault-based testing approach, when applied to the Openpilot case study, successfully generated mutations, providing specific insights into the effectiveness of the implemented test suite.

With the help of our approach, engineers or developers build system models by focusing on safety concerns. They use these models as an input to mutation testing and evaluate the adequacy of the test suite based on the safety concerns explicitly defined in the models. If the mutation score is not 100, they revisit and reiterate the test suite to add missing or edge test cases to achieve a mutation score of 100. Our tool automates the process by removing manual steps for generating the mutations and running test cases. It also helps to ensure that safety concerns are properly addressed in the test cases by focusing on safety tactics.

8 DSL evaluation

In this section, we present the evaluation of our DSL from the end users' perspective. Since our DSL is relatively new, we do not have adequate trained users to conduct formal interviews with questionnaires to evaluate our DSL. In this regard, we have looked at the existing studies in the literature to provide an approach for assessing DSLs from various perspectives. [29-32] propose different approaches to evaluate novel DSLs. For our DSL, we used Framework for Qualitative Assessment of DSLs (FQAD) [30], which is based on the ISO/IEC 25010:2011 standard. FQAD describes a set of quality properties for assessing a DSL, including Functional suitability, Usability, Reliability, Maintainability, Productivity, Extensibility, Compatibility, Expressiveness, Reusability, and Integrability. In the following, we present the evaluation of our DSL considering each quality characteristic.

8.1 Functional suitability

Functional suitability indicates to what degree the DSL is fully developed. This means that all necessary functionalities exist in the DSL, and the DSL does not have functionality not given in the represented domain. We used our DSL to define multiple case studies, and we have been able to describe all the problem-specific functionalities needed to express safety. From this point, we can conclude that our DSL meets this criterion.

8.2 Usability

Usability refers to the degree to which specified users can use DSL to accomplish specified goals. To analyze this property, we have asked engineers experienced in the safety domain to assess the overall usability of our DSL. We conducted usability assessments with a diverse group of 10 experienced engineers including individuals with backgrounds in software engineering and safety analysis. Each engineer dedicated an average of 12 h to thoroughly evaluate the overall usability of our DSL. Their feedback highlighted the effectiveness of the DSL. They expressed that differentiating between safety-critical and non-safetycritical components in the system helped them identify where to focus on the safety requirements. They also indicated that expressing and seeing the direct relation between safety-critical components and safety tactics helped address safety concerns in the system. Overall, they mentioned that the DSL is easy to learn and use.

8.3 Reliability

Reliability of a DSL is defined as the property of a language that helps to produce reliable programs. We developed our DSL using the Xtext framework in the Eclipse environment. The Xtext framework provides full infrastructure including parser, linker, type checker, compiler and editing support for Eclipse. The Eclipse editor provides all the requirements for handling code errors.

8.4 Maintainability

Maintainability shows to what degree the DSL is easy to maintain. Our DSL consists of four main parts, which are defined by applying the separation of concerns principle. This helps to achieve modularity in the DSL. For maintainability, it is also vital to address understandability. In our DSL, we directly model the concepts as defined in the safety domain. Therefore, the grammar is easy to understand. Maintenance also covers modifiability. Since our DSL design is modular, it can be easily modified, or new concepts can be added.

8.5 Productivity

Productivity refers to the degree to which a DSL promotes programming productivity. Our DSL helps to increase productivity because it enhances the design and testing process of safety-critical systems. It helps developers and engineers to identify safety-critical concerns by explicitly defining them at the early stages of the design. Also, it supports the testing stage by helping test engineers to assess the quality of the test suites focusing on the safety concerns.

8.6 Extensibility

Extensibility defines the degree to which a DSL has general mechanisms for users to add new features. Our DSL can easily be extended because of its modularity. Our DSL consists of four different parts that each provides different a viewpoint to the safety domain. In this regard, our DSL can be easily extended by adding new concepts. Also, the Xtext framework and the Eclipse helps users to add new features to DSL easily.

8.7 Compatibility

Compatibility of a DSL shows at what degree a DSL is compatible with the domain and the development process.

We defined our DSL to enhance the testing process of safety-critical systems. It is designed to help test engineers to assess the test suites' quality by focusing on safety concerns. It fits the systems engineering lifecycle in terms of requirement analysis, design, development, and testing.

8.8 Expressiveness

Expressiveness defines the relation between the program and what the programmer has in their mind. For this criterion, it is imperative to have a one-to-one mapping between the concepts and their representation in the DSL. We developed our DSL based on a thorough domain analysis whereby we have modeled each concept in the corresponding metamodel of the language. We can affirm a one-to-one correspondence between the concepts, and their representation in the DSL and there are no duplicated concepts. We also considered the abstraction level of the concepts in the DSL to ensure that they are not too generic or too specific but expressive enough to represent the safety domain.

8.9 Reusability

Reusability of a DSL refers to the degree to which DSL can be used in any other language. The definitions in our DSL can be used in any other language since the DSL directly models the concepts as defined in the safety domain.

8.10 Integrability

Integrability defines the degree to which the DSL is compatible with integration with other languages. We developed the DSL using the Xtext framework in Eclipse environment. The Eclipse platform allows developers to extend Eclipse applications like Eclipse IDE with additional functionalities via Eclipse plug-ins. In this respect, our DSL can be integrated with other languages using the Eclipse IDE.

9 Related work

Several studies have proposed domain-specific languages (DSL) for addressing safety. In [33], the authors define a DSL to present Petri-nets and a tool MeeNET to debug safety-critical systems. Their focus is on having a DSL to formally define behavior of the system using Petri-nets and verify the system behaviour. Nandi et al. [34] propose a DSL for the correct deployment of RV solutions in the scope of cyber-physical systems. Kaleeswaran et al. [35] define a DSL for Hazard and Operability Analysis

(HAZOP) study. HAZOP study is a systematic way to identify potential hazards in the system. The HAZOP-DSL helps users to build links between HAZOP study and the system model for consistency and traceability. With the DSL support, the changes in the system model can be detected, and the user is enforced to make necessary changes in the HAZOP. Their study enables users to detect issues in the safety analysis at early design levels. [36] defines a DSL for defining the safety requirements and automatically verifying their consistency using formal methods. They propose a domain-specific language SafeNL to enable users to define safety requirements formally. They convert the SafeNL documents into formal constraints (Clock Constraint Specification Language) and verify their consistency with existing tools. Queiroz et al. [37] propose a DSL for defining scenarios used in simulation testing of autonomous driving systems. Their DSL includes information about vehicles, pedestrians, paths, roads which are the main components to compose test cases for self-driving vehicle testing. Some studies [38, 39] define a DSL to improve processes in their engineering life cycle. [39] defines a DSL (Mauve) for specifying the software architecture of autonomous robots. Using this DSL, they analyze the real-time correctness of the architecture by verifying the schedulability of different components. They transform Mauve model into Periodic State Machines and analyze real-time characteristics of the architectural components. Also, they check the validity of behavioral properties by converting Mavue model into Fiarce [40] models and analyze the Fiarce model using TINA [41]. In [42], the authors propose a domain-specific modeling language (DSML) to provide a conceptual model for expressing the information mandated by DO-178C standard. Iber et al. [43] proposes a DSL to specify tests from UML Testing Profile (UTP). They model the UTP to support model-driven development processes such as generating test code.

All these and other studies in the literature define DSL to support system development lifecycle. Some focus only on one step of the lifecycle such as requirement analysis or design or implementation, where some of them are specific to one safety-critical domain like railway systems, robotics, or automotive. To the best of our knowledge, no generic DSL has been presented which is dedicated to assess the quality of the test suite. With this study, our primary focus is evaluating the adequacy of the test suite based on the applied safety tactics in the software architecture models. Our DSL allows us to express safety tactics in the safety models and the system implementation details to generate mutants and run test cases.

In the literature, several studies proposed a fault-based testing approach to test safety-critical systems. In [44], the authors proposes an approach to generate test oracles from

the formal requirements defined in CASDL (Casco Accurate Specification Description Language). In another study [12], a test case generation approach is defined based on model mutation for the safety requirements in the system. Firstly, a fault model is defined by describing mutation operators and UML models of the system. Then, they define a process for transforming a UML model to OOAS (Object-Oriented Action Systems) using fault models. Subsequently, OOAS models' mutations are generated and used for the test case generation process. Another study [45] applies mutation testing on a nuclear reactor. In this work, a test case generation approach is defined to test a nuclear reactor. Mutation testing is applied by mutating the source code. With this approach, they aim to calculate the degree of test adequacy of the generated test cases.

Safety concern has not been explicitly addressed using a dedicated architecture perspective before. However, there is plenty of work related to safety engineering. In our earlier work [19, 46], we have provided a safety perspective that can support the architectural design of safety-critical systems. It can assist the system and software architects in designing, analyzing, and communicating the decisions regarding safety concerns by evaluating safety issues early on the life cycle before implementing the system.

In [47, 48], several architectural patterns are proposed to support software safety design. Gawand et al. [48] propose a framework for the specification of architectural patterns to support safety and fault tolerance. They provide four types of patterns. One of the patterns is Control-Monitor pattern. They aim to improve fault detection by using redundancy by using this pattern. Another pattern is the Triple Modular Redundancy pattern which is used to enhance system's safety where there is no failsafe state. The other pattern is the Reflective State pattern which separates the application into base-level and meta-level to separate control and safety aspects from the application logic. The last pattern is Fault Tolerance Redundancy pattern which improves the fault tolerance of the system while implementing the redundancy for safety. Armoush et al. [47] propose a Recovery Block with Backup Voting pattern which improves the fault tolerance of the system.

Our earlier work considered the explicit modeling of viewpoints for quality concerns [49–51]. As a result, each quality concern, such as adaptability and recoverability, requires a different decomposition of the architecture. Architectural elements and relations are defined to specify the required decomposition for the quality concerns. Earlier work on local recoverability has shown that this approach is also broadly applicable. We consider this work complementary to the architectural perspectives approach. Both alternative approaches seem to have merits.

10 Threats to validity

In this section, we discuss threats to the validity of our study using the guideline defined in [52].

10.1 Construct validity

The main goal of our study is to assess the adequacy of the test suite of safety-critical systems. To achieve this, we built an approach by leveraging existing fault-based testing methods. While applying the proposed approach to the case study, we assumed that the implementation (code) of the case study is bug-free and the test suite is complete from a test coverage perspective. Any defects in the case study implementation and test suite may affect our case study evaluation results. Additionally, we used μ Java [53] to generate mutants. Any issues in μ Java would jeopardize our study's construct validity. The other point we bring to attention is that our case study is implemented in Java. We did not focus on the requirements of programming languages for supporting the implementation of safety-critical concerns since the scope of the paper is on mutant generation. This also might have an effect on the construct validity.

10.2 Internal validity

To evaluate our approach, we used a use case from a real industrial case study. For the given case study, there is no equivalent mutant detected. Some of the equivalent mutants could not be generated or found because of the size of the case study. This might cause a threat to our internal validity. We plan to perform additional case studies as future work.

10.3 External validity

Our approach is based on safety concerns, and it provides a generic approach for safety-critical systems. To illustrate our work, we applied our approach to a real case from a safety-critical system in the avionics domain. However, it can be applied to any safety-critical system from any domain since the overall approach is generic.

10.4 Reliability

In this work, we provided detailed information about each step for the proposed approach. The metamodel, DSL, and tool are publicly available through an open-source platform. Hence, the results obtained within our study are reproducible.

11 Conclusion

Testing safety-critical systems is essential and for this purpose, developing an effective test suite is necessary. In this article, we have thus provided a systematic approach for assessing test suites of safety-critical systems. For this purpose, we have adopted a fault-based testing approach that can be used to analyze the effectiveness of so-called architecture safety tactics. We have developed the required metamodel and realized the DSL to model the faults and tactics and support fault-based testing. We have applied the approach and the tool for a real industrial case study. The approach and the tool are helpful to assess a given test suite and analyze the strength of the safety tactics.

Based on the results from our case study, our main conclusion is that our approach is feasible and effective for test suite assessment of safety-critical systems. It supports the overall architecture design of safety-critical systems and analysis to realize the requirements for safety-critical systems. The iterative process facilitated by our approach ensures that safety concerns are thoroughly addressed in the test cases, supported by the automation provided by our tool in mutation generation and test case execution. The evaluation of our fault-based testing approach revealed the successful generation of mutations for the case study implementation. While some safety requirements achieved a 100% mutation score, highlighting the effectiveness of the implemented test suite, others, such as SR6, identified the need for additional test cases. These findings support the effectiveness of our approach for safety-critical systems, offering insights for enhancing systems like openpilot in the realm of autonomous driving.

With our fault-based testing approach, engineers and developers build dedicated system models to express safety concerns, use these models as an input to mutation testing, and evaluate the effectiveness of the developed test suite based on the safety concerns addressed in the models. If the mutation score is not 100, they revisit and reiterate the test suite to add missing or edge test cases to achieve a mutation score of 100. Since our approach focuses on safety tactics and fault knowledge, it enables developers to build complete and robust test suites focusing on safety concerns while building safety-critical systems.

For future developments, we aim to enhance the approach further by systematically analyzing different faults and safety tactics from various domains such as robotics, nuclear systems, and automotive. Besides, we aim to automate mutation operation selection with the help of the safety model. Another improvement area that we plan is generating test artifacts (test data, test scripts, test oracle) from the DSL we defined. We also consider adding debugging and testing support to our DSL as future work.

Appendix: EBNF grammar of DSL

```
SafetyDSL = {SafetyView} ImplementationDetail;
SafetyView = HazardView | SafetyTacticView | SafetyCriticalView;
HazardView = 'Hazard View' STRING '{ Elements {' {HazardElement} '}' 'Relations {' {HazardRelation} '}' '}' ;
HazardElement = Hazard | SafetyRequirement | Consequence | Fault | FaultTree ;
Hazard = Imaard | SafetyRequirement | Consequence | Fault | 'fault 
FaultTree = 'faultTree' FaultTreeID FaultTreeNode ';' ;
ANDNode = FaultID | ANDNode | ORNode ;
ANDNode = FaultTreeNode 'AND' FaultTreeNode ;
ORNode = FaultTreeNode 'OR' FaultTreeNode
HazardRelation = DerivedFrom | Causes | CausedBy ;
DerivedFrom = SReqID {', 'SReqID} 'derivedFrom' HazardID ';' ;
Causes = HazardID 'causes' ConsequenceID {',' ConsequenceID} ';' ;
CausedBy = HazardID 'causedBy' FaultTreeID ';' ;
SafetyTacticView= 'SafetyTacticView' STRING '{' {SafetyTactic} '}';
SafetyTactic = ('faultAvoidance' | 'faultDetection' | 'faultContainment') STacticID '{' 'type=' STRING 'handledFaults=
(FaultID) {',' FaultID} '}' ';';
SafetyCriticalView = 'Safety-CriticalView' name=ID '{' 'Elements {'{ArchitecturalElement} '}' 'Relations
{'{SafetyCriticalRelation} '}' '}';
{ {safetyCriticalRelation } ; ;
ArchitecturalElement = SafetyCritical | NonSafetyCritical | Monitor ;
SafetyCritical = 'safety-critical' SCModuleID '{' 'criticalityLevel=' ('A' | 'B' | 'C' | 'D') ';'
'implementedSafetyRequirements=' SReqID {',' SReqID} ';'
['implementedTactics=' STacticID {',' STacticID} ';' ]
['sub-elements='{SCModuleID} {',' SCModuleID} ';' ]
                   ['hasState' StateID {',' StateID}] '}' ';
NonSafetyCritical = 'non-safety-critical' NSCModuleID ( '{' {NSCModuleID} '}' | ';' );
Monitor = 'monitor' MID ['{' 'implementedTactics=' SCTacticID {',' SCTacticID} '}'] ';';
State = ('state' | 'safeState') StateID ';';
SafetyCriticalRelation = ArchElementToArchElement |
                                                                                                                                MonitorToArchElement | ReportsFault ;
ArchitecuralElementID = (SCModuleID | NSCModuleID |
ArchitecuralElementID = (SCModuleID | NSCModuleID | MID);
ArchElementToArchElement = ArchElementID ('reads' | 'writes' | 'commands') ArchElementID {',' ArchElementID } ';' ;
MonitorToArchElement = MID ('stops' | 'starts' | 'inits' | 'restarts' | 'monitors') SCModuleID {';' SCModuleID} ';'
ReportsFault = SCModuleID 'reportsFault' SCModuleID {',' SCModuleID} ';' ;
ImplementationDetail = 'ImplementationRelations {' 'Module-Class Relation {' {ModuleClassRelation} '}:'
                'Class-TestCase Relation{' {ClassTestCaseRelation}
                                                                                                                                                   '};'
ModuleClassRelation = SCModuleID 'composesOf=' '{' ClassDef {',' ClassDef} '}' ';' ;
TacticTestCaseRelation = STacticID 'testWith=' '{' QualifiedName {',' QualifiedName
                                                                                                                                 QualifiedName {',' QualifiedName} '}' ';' ;
ClassDef = QualifiedName ;
Qualified Name = STRING { '.' STRING } ;
                                                                                                                                 FaultTreeID = STRING;
                                                                                                                                 SCModuleID = STRING;
HazardID = STRING:
                                                                                                                                 NSCModuleID = STRING:
SReqID = STRING;
                                                                                                                                 MID = STRING;
ConsequenceID = STRING;
                                                                                                                                 StateID = STRING:
FaultID = STRING;
                                                                                                                                 STacticID = STRING;
```

Author contributions HGG: Conceptualization, Methodology, Software, Writing-Original Draft, Writing-Review & Editing Bedir Tekinerdogan: Conceptualization, Methodology, Software, Writing-Original Draft, Writing-Review & Editing, Supervision CC: Conceptualization, Methodology, Software, Writing-Original Draft, Writing-Review & Editing, Supervision NaPE: Conceptualization, Writing-Review & Editing.

Funding No funding was received for conducting this study.

Data availability Enquiries about data availability should be directed to the authors.

Declarations

Competing interests The authors have no competing interest to declare that are relevant to the content of this article.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Leveson, N., Harvey, P.: Analyzing software safety. IEEE Trans. Softw. Eng. 9(05), 569–579 (1983). https://doi.org/10.1109/TSE. 1983.235116
- Ericson, C.: Hazard Analysis Techniques for System Safety. Wiley (2005). https://doi.org/10.1002/0471739421.fmatter
- Sparkman, D.: Techniques, processes, and measures for software safety and reliability., Nuclear Systems Safety Program https:// doi.org/10.2172/6801101
- Leveson, N.G.: Safeware: System Safety and Computers, ACM, New York (1995). https://doi.org/10.1145/202709
- NASA Software Safety Guidebook, URL: https://standards.nasa. gov/standard/nasa/nasa-gb-871913
- Rozanski, N., Woods, E.: Software Systems Architecture: Working With Stakeholders Using Viewpoints and Perspectives. Addison-Wesley Professional (2005). https://doi.org/10.5555/ 1036701
- Wu, W., Kelly, T.: Safety tactics for software architecture design, in: Proceedings of the 28th Annual International Computer Software and Applications Conference, 2004. COMPSAC 2004., 2004, pp. 368–375 vol.1. https://doi.org/10.1109/CMPSAC.2004. 1342860
- Gurbuz, H.G., Pala Er N., Tekinerdogan, B.: Architecture framework for software safety, in: System Analysis and Modeling: Models and Reusability, Springer International Publishing, Cham, 2014, pp. 64–79. https://doi.org/10.1007/978-3-319-11743-0_5
- Xue-Fang, D., Rui, Z.: A scenario-based lightweight software architecture analysis method, in: 3rd International Conference on Green Communications and Networks, Vol. 54 Volume 1, 2014, pp. 949–956. https://doi.org/10.2495/GCN131272
- Buchgeher, G., Weinreich, R.: An approach for combining model-based and scenario-based software architecture analysis, in. Fifth International Conference on Software Engineering Advances 2010, 141–148 (2010). https://doi.org/10.1109/ICSEA. 2010.29
- Tekinerdogan, B., Sozer, H., Aksit, M.: Software architecture reliability analysis using failure scenarios. J. Syst. Softw. 81(4), 558–575 (2008). https://doi.org/10.1016/j.jss.2007.10.029
- Herzner, W., Schlick, R., Brandl, H., Wiessalla, J.: Towards fault-based generation of test cases for dependable embedded software, Softwaretechnik-Trends 31
- Preschern, C., Kajtazovic, N., Kreiner, C., et al., Catalog of safety tactics in the light of the IEC 61508 safety lifecycle, in: Proceedings of VikingPLoP 2013 Conference, (2013), p. 79
- 14. Pezze, M., Young, M.: Software Testing and Analysis: Process, Principles, and Techniques, Wiley, (2007)
- Ammann, P., Offutt, J.: Introduction to Software Testing, 1st edn. Cambridge University Press, Cambridge (2008)
- 16. Openpilot: an open-source driver assistance system., URL: https://github.com/commaai/openpilot
- 17. Galanopoulou, R., Spinellis, D.D.: A dataset of open-source safety-critical software, (2021)
- 18. Openpilot high-level component diagram., URL: https://github. com/commaai/openpilot/wiki/Introduction-to-openpilot#inter-pro cess-communication

- Gurbuz, H.G., Tekinerdogan B., Pala Er, N.: Safety perspective for supporting architectural design of safety-critical systems, in: Software Architecture, Springer International Publishing, Cham, (2014), pp. 365–373. https://doi.org/10.1007/978-3-319-09970-5_31
- 20. ISO/IEC 14977:1996 information technology syntactic metalanguage - extended bnf (1996)
- [MIL-STD-882D], Standard Practice for System Safety, Department of Defense (2000)
- 22. Mutpy, mutation testing tool for python 3.3, URL: https://github. com/mutpy/mutpy
- 23. Xtext home page, URL: https://www.eclipse.org/Xtext
- 24. Xtend home page, URL: https://www.eclipse.org/xtend
- 25. Safety DSL tool, URL: http://github.com/havvagulay/safetyDsl
- 26. A. J. Offutt, Jie Pan, Detecting equivalent mutants and the feasible path problem, in: Proceedings of 11th Annual Conference on Computer Assurance. COMPASS '96, 1996, pp. 224–236. https://doi.org/10.1109/CMPASS.1996.507890
- Madeyski, L., Orzeszyna, W., Torkar, R., Józala, M.: Overcoming the equivalent mutant problem: a systematic literature review and a comparative experiment of second order mutation. IEEE Trans. Softw. Eng. 40(1), 23–42 (2014). https://doi.org/10.1109/ TSE.2013.44
- Adamopoulos, K., Harman, M., Hierons, R.: How to overcome the equivalent mutant problem and achieve tailored selective mutation using co-evolution. Genet. and Evol. Comput. 3103(2004), 1338–1349 (2004). https://doi.org/10.1007/978-3-540-24855-2_155
- Mernik, M., Heering, J., Sloane, A.: When and how to develop domain-specific languages. ACM Comput. Surv. 37, 316–344 (2005). https://doi.org/10.1145/1118890.1118892
- Kahraman, G., Bilgen, S.: Aframework for qualitative assessment of domain-specific languages. Softw. Syst. Model. 14(4), 1505–1526 (2015). https://doi.org/10.1007/s10270-013-0387-8
- N. Oliveira, M. J. V. Pereira, P. R. Henriques, D. C. da Cruz, Domain specific languages: a theoretical survey, in: INForum'09-Simposio de Informatica, (2009)
- Kosar, T., Oliveira, N., Mernik, M., João, M., Pereira, M., Repinšek, M., Cruz, D., Rangel Henriques, P.: Comparing general-purpose and domain-specific languages: an empirical study. Comput. Sci. Inf. Syst. (2010). https://doi.org/10.2298/ CSIS1002247K
- Idani, A.: Formal model-driven executable DSLs. Innov. Syst. Softw. Eng. (2022). https://doi.org/10.1007/s11334-021-00408-4
- Nandi, G.S., Pereira, D., Proença, J., Tovar, E.: Work-in-progress: a dsl for the safe deployment of runtime monitors in cyberphysical systems, in. IEEE Real-Time Systems Symposium (RTSS) 2020, 395–398 (2020). https://doi.org/10.1109/ RTSS49844.2020.00047
- Kaleeswaran, A.P., Munk, P., Sarkic, S., Vogel, T., Nordmann, A.: A domain specific language to support HAZOP studies of SysML models. In: Papadopoulos, Y., Aslansefat, K., Katsaros, P., Bozzano, M. (eds.) Model-Based Safety and Assessment, pp. 47–62. Springer International Publishing, Cham (2019). https://doi.org/10.1007/978-3-030-32872-6_4
- Chen, X., Zhong, Z., Jin, Z., Zhang, M., Li T., Chen, X., Zhou, T.: Automating consistency verification of safety requirements for railway interlocking systems, in: 2019 IEEE 27th International Requirements Engineering Conference (RE), 2019, pp. 308–318. https://doi.org/10.1109/RE.2019.00040
- Queiroz, R., Berger, T., Czarnecki, K.: Geoscenario: An open DSL for autonomous driving scenario representation, in. IEEE Intelligent Vehicles Symposium (IV) 2019, 287–294 (2019). https://doi.org/10.1109/IVS.2019.8814107
- 38. Anderson, M., Bowman, J., Kilgo, P.: Rdis: Generalizing domain concepts to specify device to framework mappings, in. IEEE

International Conference on Robotics and Automation 2012, 1836–1841 (2012). https://doi.org/10.1109/ICRA.2012.6225197

- Gobillot, N., Lesire, C., Doose, D.: A modeling framework for software architecture specification and validation, in: Simulation, Modeling, and Programming for Autonomous Robots, 2014, pp. 303–314. https://doi.org/10.1007/978-3-319-11900-7_26
- 40. Berthomieu, B., Bodeveix, J.-p., Farail, P., Filali, M., Garavel, H., Gaufillet, P., Lang F., F. Vernadat, Fiacre: an intermediate language for model verification in the topcased environment, Proc. of the Embedded Real Time Software (ERTS) p. 8. (2008)
- B. Berthomieu, F. Vernadat, Time petri nets analysis with tina, 2006, pp. 123–124. https://doi.org/10.1109/QEST.2006.56
- 42. Metayer, N., Paz, A., El Boussaidi, G.: Modelling do-178c assurance needs: A design assurance level-sensitive dsl, in. IEEE International Symposium on Software Reliability Engineering Workshops (ISSREW) 2019, 338–345 (2019). https://doi.org/10. 1109/ISSREW.2019.00094
- J. Iber, N. Kajtazović, A. Höller, T. Rauter, C. Kreiner, Ubtl uml testing profile based testing language, in: 2015 3rd International Conference on Model-Driven Engineering and Software Development (MODELSWARD), 2015, pp. 1–12. https://doi.org/10. 5220/0005241300990110
- 44. Zheng, H., Feng, J., Miao, W., Pu, G.: Generating test cases from requirements: A case study in railway control system domain, in. International Symposium on Theoretical Aspects of Software Engineering (TASE) 2021, 183–190 (2021). https://doi.org/10. 1109/TASE52547.2021.00029
- Babu, P.A., Kumar, C.S., Murali, N., Jayakumar, T.: An intuitive approach to determine test adequacy in safety-critical software. SIGSOFT Softw. Eng. Notes 37(5), 1–10 (2012). https://doi.org/ 10.1145/2347696.2347701
- Gurbuz, H.G., Tekinerdogan, B.: Model-based testing for software safety: a systematic mapping study. Softw. Qual. J. 26(4), 1327–1372 (2018). https://doi.org/10.1007/s11219-017-9386-2
- Armoush, A., Salewski, F., Kowalewski, S.: Recovery block with backup voting: A new pattern with extended representation for safety critical embedded systems, in. International Conference on Information Technology 2008, 232–237 (2008). https://doi.org/ 10.1109/ICIT.2008.60
- Gawand, H., Mundada, R.S., Swaminathan, P.: Design patterns to implement safety and fault tolerance. Int. J. Comput. Appl. 18, 6–13 (2011). https://doi.org/10.5120/2259-2900
- H. Sozer, B. Tekinerdogan, Introducing recovery style for modeling and analyzing system recovery, in: Seventh Working IEEE/ IFIP Conference on Software Architecture (WICSA 2008), 2008, pp. 167–176. https://doi.org/10.1109/WICSA.2008.9
- Sozer, H., Tekinerdoğan, B., Akşit, M.: Optimizing decomposition of software architecture for local recovery. Softw. Qual. J. 21(2), 203–240 (2013). https://doi.org/10.1007/s11219-011-9171-6
- Tekinerdogan, B., Sozer, H.: Defining architectural viewpoints for quality concerns. Softw. Archit. 6903, 26–34 (2011). https:// doi.org/10.1007/978-3-642-23798-0_3

- Runeson, P., Höst, M.: Guidelines for conducting and reporting case study research in software engineering. Empir. Softw. Eng. 14(2), 131 (2008). https://doi.org/10.1007/s10664-008-9102-8
- mujava home page, URL: http://cs.gmu.edu/~offutt/mujava (2014)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Havva Gulay Gurbuz is PhD candidate at Information Technology group at Wageningen University & Research (WUR) in the Netherlands and a software engineer manager at Microsoft USA. She received MSc degree (2014) in Computer Engineering from Bilkent University, Turkey and B.S. degree (2012) in Computer Engineering from Hacettepe University, Turkey.



Professor Tekinerdogan is a computer scientist with over 25 years of experience in software/ systems engineering and information technology. Throughout his career, he has authored more than 400 peer-reviewed scientific papers and 8 edited books. He has been involved in dozens of national and international research and consultancy projects with various large software companies as a principal researcher and leading software/ system architect. His experience

spans a broad range of domains, including consumer electronics, enterprise systems, automotive systems, critical infrastructures, cyber-physical systems, satellite systems, defense systems, production line systems, command and control systems, physical protection systems, radar systems, smart metering systems, energy systems, and precision farming. He takes a holistic, systemic, and interdisciplinary approach to solving real industrial problems and has ample experience in software and systems architecting, software and systems product line engineering, cyber-physical systems, model-driven software engineering, aspect-oriented software engineering, global software development, systems engineering, system of systems engineering, data science, and artificial intelligence. In addition to his research, he is an active educator. He has developed and taught around 20 different academic courses and provided many software/ systems engineering courses to more than 50 companies in The Netherlands, Germany, India, and Turkey.



Cagatay Catal is a Full Professor at Qatar University, Department of Computer Science & Engineering. He obtained his BSc and MSc degrees in Computer Engineering from Istanbul Technical University in 2002 and 2004, respectively. He later pursued his Ph.D. in Computer Engineering at Yildiz Technical University in Istanbul, completing it in 2008. From 2020 to 2021, he held the position of Full Professor in the Department of Computer Engineering

at Bahcesehir University in Istanbul. Prior to that, he worked two years as a full-time faculty member at Wageningen University & Research (WUR) in the Netherlands. Before his time at WUR, Dr. Catal worked for six years in the Department of Computer Engineering at Istanbul Kultur University, where he served as an Associate Professor and Head of the Department. In April 2014, he was awarded the title of Associate Professor by the Inter-University Council of Turkey. Before entering the academic field, Dr. Catal had 8 years of experience at the Scientific and Technological Research Council of Turkey (TUBITAK), Information Technologies Institute, where he held the positions of Senior Researcher, Project Manager, and Senior Software Engineer. His research interests encompass various areas, including Artificial Intelligence, Deep Learning, Machine Learning, Natural Language Processing, Software Engineering, Software Testing, Software Architecture, and Precision Agriculture.



Nagehan Pala Er has bachelor's degree in Software Engineering of Hacettepe University and master's degree in the same field of Bilkent University. She has been working at ASELSAN, which is a leading company in defence industry in Turkey, for more than 15 years. She has been involved in architecture and implementation of safety critical software intensive systems in avionics domain.